Author's response to reviews

Title: Preoperative workup in the assessment of adrenal incidentalomas. Outcome from 506 consecutive laparoscopic adrenalectomies.

Authors:

Mario Musella (mario.musella@unina.it)
Giovanni Conzo (giovanni.conzo@unina2.it)
Marco Milone (milone.marco@alice.it)
Francesco Corcione (francesco.corcione@ospedalideicolli.it)
Giulio Belli (chirurgia.loretonuovo@tin.it)
Maurizio De Palma (mauridep@tin.it)
Annunziato Tricarico (annunziato.tricarico@tiscali.it)
Luigi Santini (luigi.santini@unina2.it)
Antonietta Palazzo (palazzo.antonella@libero.it)
Paolo Bianco (biancopaolo@virgilio.it)
Bernadette Biondi (biondi@unina.it)
Rosario Pivonello (rosario.pivonello@unina.it)
Annamaria Colao (colao@unina.it)

Version: 3 Date: 14 June 2013

Author's response to reviews: see over
Dear Ms. Costoy, as requested we provided the following changes. 
In the Methods section (rows 18-22) it is now stated what you suggested.

1. The use of all data concerning this retrospective study was authorized by the ethical committee of the “Federico II” University. This is a default authorization, following a request, in retrospective study. Furthermore it must be considered that the “Federico II” University ethical committee is responsible for all ethic issues in our region, in this light it covers all the centres working in Campania.

2. The informed consent that all patients sign before surgery is requested mandatorily by Italian laws, and in our case, it is periodically checked and waived by our ethical committee according to the ever changing, different, medical or surgical procedures.

Yours Sincerely

Mario Musella M.D.
Associate Professor of Surgery
Naples "Federico II" University
Advanced Biomedical Sciences Department
AOU "Federico II" - Via S. Pansini 5 Building 12
80131 - Naples - Italy
Tel ++390817462880
Fax ++390817462896
mario.musella@unina.it